

Paper: **“Transformative actions from learning assessment:
A cross-context typology, coding protocol, and baseline evidence from a public university”**

Corresponding Author: **Hector A. Aponte-Alequin, PhD**

Doi: 10.19044/ejes.v12no2a1

Peer review:

Reviewer 1: Dr. Jorge Balladares, Ecuador

Reviewer 2: Blinded